Author's response to reviews

Title: Design, rationale, and baseline characteristics of a cluster randomized controlled trial of pay for performance for hypertension treatment

Authors:

Laura A Petersen (laurap@bcm.tmc.edu)
Tracy Urech (tracy.urech@va.gov)
Kate Simpson (kate.simpson@va.gov)
Kenneth Pietz (pietz.kennethc@va.gov)
Sylvia J Hysong (sylvia.hysong@va.gov)
Jochen Profit (profit@bcm.tmc.edu)
Douglas Conrad (dconrad@u.washington.edu)
R. Adams Dudley (adams.dudley@ucsf.edu)
Meghan Z Lutschg (meghan.zimmer@va.gov)
Robert Petzel (robert.petzel@va.gov)
LeChauncy D Woodard (lwoodard@bcm.tmc.edu)

Version: 4 Date: 21 September 2011

Author's response to reviews: see over
September 21, 2011

Editors

Implementation Science

c/o BioMed Central
236 Gray’s Inn Road
London WC1X 8HB
United Kingdom

Dear Editors,

We received the copy-edited version of the manuscript (MS: 1728409500580072) and have accepted all the proposed changes by the copy-editor. We have uploaded a clean version of the manuscript to the journal’s website.

Best regards,

Laura A. Petersen, MD, MPH
Professor of Medicine
Chief, Section of Health Services Research
Baylor College of Medicine

Michael E. DeBakey VA Medical Center